Preliminaries

Testing conditions unchanged w.r.t. last presentation

• SAM Dataset Definition: reco p11.09.00 lancs ttbar 0mb • Executable: bcJetAnalyze x t02.33.00 (w/ minor patch): – Protect against crash in eTag recomputations (crashes in

EMparticle::road eop() if road had no energy. Still not fixed in t02.35.00??

• Comparison:

1. Run bcjet analyze directly on reco Chunks (black) 2. Drop reco Chunks, unpack ThumbNailChunk, run bcjet analyze (dashed, blue) • Compare Block by Block (bcJet & individual taggers) Limitations on # lepton tags / jet: robust solution possible, but haven’t yet had the time to implement it

• Focus on problems rather than items addressed already • Also look more closely at precision

Frank Filthaut b-ID ThumbNail tests 24 August 2002

eTag

Frank Filthaut b-ID ThumbNail tests 24 August 2002

eTag

Frank Filthaut b-ID ThumbNail tests 24 August 2002

eTag

Frank Filthaut b-ID ThumbNail tests 24 August 2002

Frank Filthaut b-ID ThumbNail tests 24 August 2002

Frank Filthaut b-ID ThumbNail tests 24 August 2002

Frank Filthaut b-ID ThumbNail tests 24 August 2002

Frank Filthaut b-ID ThumbNail tests 24 August 2002

likeTag

Frank Filthaut b-ID ThumbNail tests 24 August 2002

Conclusions

Status:

• bcJet, vertex tag, NN tag Blocks OK • Likelihood tag Block needs checking (presumably track ↔ Vertex

association) • Electron tag variables: ΔR, p/E • Impact parameter tag track variables: ϕ, DCA (z)

Plan of action:

• Implement proper way to deal with limitations on # lepton tags, tracks /

bcJet (requires modifications to thumbnail as well as bcjet evt

packages) • Need this is also for track ↔ Jet association if we are ever to be able to run the impact parameter tag off the Thumbnail • Insufficient precision in variables that would be needed for redoing impact parameter tag (will find out which)

Frank Filthaut b-ID ThumbNail tests 24 August 2002